Author's response to reviews

Title: Ultra-early versus early salvage androgen deprivation therapy for post-prostatectomy biochemical recurrence in pT2-4N0M0 prostate cancer

Authors:

Satoru Taguchi (satorutaguchi33@gmail.com)
Hiroshi Fukuhara (hfukuhara-jua@umin.ac.jp)
Takeshi Azuma (tazuma-tky@umin.ac.jp)
Motofumi Suzuki (suzukim-uro@h.u-tokyo.ac.jp)
Tetsuya Fujimura (FUJIMURAT-URQ@h.u-tokyo.ac.jp)
Tohru Nakagawa (tohru-tky@umin.ac.jp)
Akira Ishikawa (beetle-55@umin.org)
Haruki Kume (kume@kuc.biglobe.ne.jp)
Yasuhiko Igawa (yigawa-jua@umin.ac.jp)
Yukio Homma (homma-uro@umin.ac.jp)

Version: 7  Date: 17 September 2014

Author's response to reviews:

I forget the change of the title in the web (not in the text attached).

I have changed the title in the web from 'Ultra-early versus early salvage hormonal therapy for post-prostatectomy biochemical recurrence in pT2-4N0M0 prostate cancer' to 'Ultra-early versus early salvage androgen deprivation therapy for post-prostatectomy biochemical recurrence in pT2-4N0M0 prostate cancer'.

Thank you in advance.